Author’s response to reviews

Title: INfluence of Successful Periodontal Intervention in Renal Disease (INSPIRED): study protocol for a randomised controlled pilot clinical trial.

Authors:

Praveen Sharma (praveen.sharma@nhs.net)
Paul Cockwell (paul.cockwell@uhb.nhs.uk)
Thomas Dietrich (t.dietrich@bham.ac.uk)
Natalie Ives (n.j.ives@bham.ac.uk)
Charles Ferro (charles.ferro@uhb.nhs.uk)
Iain Chapple (i.l.c.chapple@bham.ac.uk)

Version: 2 Date: 02 Oct 2017

Author’s response to reviews:

We apologise for this oversight. The text has now been amended to read:

“The differences between the arms in the means and mean change from baseline to each time-point will be calculated, along with the 95% confidence intervals. For dichotomous variables, changes in proportions, instead of means, will be analysed over time. This will help determine the sensitivity of the outcome measures, such as eGFR, ACR, PWV, and measures of inflammation or oxidative stress to change following periodontal therapy”